Author's response to reviews

Title: The Development of a Web- and a Print-Based Decision Aid for Prostate Cancer Screening

Authors:

Caroline S Dorfman (csd9@georgetown.edu)
Randi M. Williams (rmw27@georgetown.edu)
Elisabeth C Kassan (eck33@georgetown.edu)
Sara N Red (snr5@georgetown.edu)
David L Dawson (dlid27@georgetown.edu)
William Tuong (wt45@georgetown.edu)
Elizabeth R Parker (erp3@georgetown.edu)
Janet Ohene-Frempong (jofrempong@comcast.net)
Kimberly M Davis (kmd54@georgetown.edu)
Alexander H Krist (ahkrist@mail1.vcu.edu)
Steven Woolf (swoolf@vcu.edu)
Marc D Schwartz (schwartm@georgetown.edu)
Mary B Fishman (mbf3@gunet.georgetown.edu)
Carmella A Cole (Carmella.a.cole@medstar.net)
Kathryn L Taylor (taylorkl@georgetown.edu)

Version: 7 Date: 27 January 2010

Author's response to reviews:

We have made all of the copyediting changes requested to the manuscript and to the tables.